Author's response to reviews

Title: Trends in sexually transmitted infections in the Netherlands, combining surveillance data from general practices and sexually transmitted infection centers

Authors:

Ingrid VF van den Broek (ingrid.van.den.broek@rivm.nl)
Robert A Verheij (R.Verheij@nivel.nl)
Christel E van Dijk (C.vanDijk@nivel.nl)
Femke DH Koedijk (femke.koedijk@rivm.nl)
Marianne AB van der Sande (Marianne.van.der.Sande@rivm.nl)
Jan EAM van Bergen (JvanBergen@soaids.nl)

Version: 3 Date: 7 May 2010

Author's response to reviews:

Dear Editor,

I have made the last changes in accordance with the suggestions in your e-mail. These were:
- Figures: figure titles are provided in the Powerpoint files.
- Renamed 'Introduction' as 'Background'.
- For authors' email addresses, a separate list on the title page was created.
- All tracked changes and comments are removed from the manuscript.
- References: all citations are placed before the full-stop; some adaptations were made in the reference list (typographic errors).
- Table description: incorporated into the table itself.
- Table: vertical lines were removed from the table.
- Figures: according to guidelines in ppt.
- Last check for typographical errors and final corrections.